Author's response to reviews

Title: Treatment with Pirfenidone for two years decreases fibrosis, cytokine levels and enhances CB2 expression in patients with chronic hepatitis C

Authors:

Lucia Flores Contreras (lucia_floresc1@yahoo.com.mx)
Ana S Sandoval Rodriguez (anasol44@hotmail.com)
Mayra G Mena Enriquez (mayra_mena7@hotmail.com)
Silvia Lucano Landeros (silvia.lucano@gmail.com)
Inmaculada Arellano Olivera (ma.inmaculadaarellanoolivera@yahoo.com.mx)
Arnulfo Alvarez Alvarez (arnulfo.alvarezalvarez@yahoo.com.mx)
Guadalupe Sanchez-Parada (sanchez.mariscal@hotmail.com)
Juan Armendariz-Borunda (armdbo@gmail.com)

Version: 7
Date: 11 July 2014

Author's response to reviews: see over
Dear Dr. Magdalena Morawska
Executive Editor
BMC Gastroenterology

Enclosed please find the revised version of our manuscript entitled “Treatment with Pirfenidone for two years decreases fibrosis, cytokine levels and enhances CB2 expression in patients with chronic hepatitis C” by L. Flores-Contreras and collaborators. It is important to notice that all changes made are in bold and underlined lettering in the new manuscript according to the reviewers’ queries. We have corrected grammar errors and added the requested point-by-point response. Additionally, we have updated reference # 23 which reflects better the scope of our manuscript.

We believe we’ve fulfilled with the reviewers’ queries and we hope that you find now our manuscript suitable for publication in your prestigious journal.

Should you need any more information on this issue, please do not hesitate to contact me.

Dr. Juan Armendariz-Borunda
Head, Department of Molecular biology and Genomics,
CUCS, University of Guadalajara,
Guadalajara, Jalisco, Mexico
Phone: (52) 33-1058-5317
E-mail: armdbo@gmail.com